Author's response to reviews

Title: Nonobstructive Hydronephrosis from Social Polydipsia: a case report.

Authors:

Natallia Maroz (natallia.maroz@gmail.com)
Uladrzimir Maroz (uladzimir.maroz@gmail.com)
Saima Iqbal (saima.iqbal@medicine.ufl.edu)
Ravi Ayer (ravi.ayer@medicine.ufl.edu)
Ganesh Kambhampati (ganesh.kambhampathi@medicine.ufl.edu)
Ahsan A Ejaz (ejazaa@medicine.ufl.edu)

Version: 3 Date: 1 October 2012

Author's response to reviews:

9/30/2012

Professor Michael Kidd AM

Editor-in-Chief

Journal of Medical Case Reports

Re: MS: 1672325195742268 - Nonobstructive hydronephrosis due to social polydipsia: a case report.

Dear Editor,

Thank you for the opportunity to submit a revised version of our manuscript titled “Nonobstructive Hydronephrosis due to Social Polydipsia: a case report.” for consideration for publication in the Journal of Medical Case Reports. We have provided a point-by-point response to Deputy Editor comments below. A “Marked Copy” highlighting the changes made is also attached with the “Revised Version.” We hope that the manuscript is now suitable for publication in the Journal.

Thank you.

Sincerely,

Natallia Maroz, M.D.

Division of Nephrology, Hypertension and Transplantation,

University of Florida,

P.O. Box 100224,
Comment #1.

There are still within the text a few grammatical and typographical errors of a minor nature.

Response: We thank the Deputy Editor for the valuable comments. The manuscript was sent to the professional copyediting service Editage for elimination of grammatical and typographical errors. Marked copy of the manuscript reflects all the changes suggested by the copyediting service. Revised copy of the manuscript is submitted.

Comment #2:

The arrows in Figure 1a to c are really hard to see. Is there anything more that can be done to make them more visible?

Response: Arrows in Figure 1a to c were replaced by different shape. The color of arrows was changed from black to white to enhance visibility.